Author’s response to reviews

Title: Assessment of Haemostasis in Patients undergoing Emergent Neurosurgery by Rotational Elastometry and Standard Coagulation Tests: A Prospective Observational Study

Authors:

Christoph Ellenberger (Christoph.Ellenberger@hcuge.ch)

Najia Garofano (Najia.PlojouxGarofano@hcuge.ch)

John Diaper (John.Diaper@hcuge.ch)

Gleicy Barcelos (Gleicy.Barcelos@hcuge.ch)

Gordana Pavlovic (Gordana.Pavlovic@hcu.ch)

Marc licker (marc-joseph.licker@hcuge.ch)

Version: 3 Date: 16 Oct 2017

Author’s response to reviews:

In this clean copy we included all editor comments:

- E-mail addresses for all of the authors in the title page
- A 'conclusions' heading in the discussion section
- A list of abbreviations
- A corrected statement regarding ethical approval and consent
- A 'funding' section together with the other declarations
- An updated authors contributions
- A figure 1 without the logo of 'University Hospital of Geneva' (the figure is your own work)
- A reference for figure 3 in the main text